Weekly Status Week of April 17 – April 23, 2008
Day of Year 108 - 114
Mission Day 2712 - 2718
Earth Observing-One (EO-1) General

There were 102 Data Collection Events (DCEs) scheduled this week.  
On April 22, a brief telecon was held with the International Federation Red Cross (IFRC) and other international organizations working in the area of worldwide flooding disaster alerts.  The following significant items were discussed:

· Principle topic was the upcoming June 6 global flood sensor web workshop to be hosted by the Space Research Institute (SRI) of National Space Agency of Ukraine (NSAU) and held in Kiev.  A preliminary list of foreign attendees has been sent to the NSAU with a finalized list due to be sent next week.  A preliminary agenda for this workshop is on the website:

http://geo-ukraine.org.ua/conference/index.php?option=com_content&task=view&id=16
INSTRUMENTS
All instruments operated nominally this week.
Instrument decontamination cycles

· ALI out-gassing was performed from April 22 (113) at 22:07z to April 23 (114) at 12:59z.

· HSI de-icing was performed from April 22 (113) at 21:55z to April 23 (104) at 12:45z.

Lunar calibration was performed on April 20, 2008.

· Nominal lunar calibration was performed during the 23:00z-23:35z umbra.

· No modified lunar calibration was performed this month.

· FOT scrubbed the execution of this month’s modified lunar calibration due to lack of knowledge of whether there is any correlation between executing the calibration and the TSM 67 being tripped.

· TSM 67 is the Telemetry Statistics Monitor that observes the value of the chassis current. 

· More information on this issue is given below under Mission Operations Center.

EO-1 Spacecraft Subsystems
Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from its Solid-State-Recorders.
Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

Rob Sohlberg and Dan Mandl presented an overview of the Fire Science Workshop that was held at the Univ. of MD in February and an overview of the Sensor Web experiences during the fall 

wildfires to the Tactical Fire Remote Sensing Advisory Committee (TFRSAC) in Salt Lake City on April 14, 2008 which was held in Salt Lake City.  Rob Sohlberg also presented “ Integrating Orbital and Airborne Assets:  SensorWeb Demonstrations During the Western State Fire Missions (WFSM) 2007” at the Remote Sensing Conference in Salt Lake City  on April 16, 2008.  Copies of the presentations and videos can be downloaded from:

http://homepage.mac.com/robert.sohlberg
A teleconference was held between the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday, April 17.  The following items of interest were discussed:

· Prasanta Bose/Ryan Timmons are to send a write-up their lightning prediction work.
· Troy Ames is to develop a Sensor Observation Service (SOS) for the OASIS Autonomous Surface Fleet.

· This summer the Ikhana Unmanned Vehicle System (UVS) will have an ocean color detection instrument installed.  This capability presents the possibility for inclusion into our estuary and coastal monitoring sensor web activities.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week with the exception of during execution of the lunar calibration.
EO-1 MISSION OPERATIONS CENTER
Real-Time
Spacecraft state of health is currently nominal.
Solar array characterization test was performed on April 23 (114) at 07:34z.  
Chassis current spikes

· FOT has noted an increase in the number of event messages for chassis current exceeding yellow limits during the past 2 months.  That is, there have been seven trips over the past 2 months as compared to seven trips over the previous 5 months.
· After each spike, TSM 67 trips and an event message is generated on-board. 

· If the value spikes over 0.2 amps at any given time the TSM first limit will be violated which generates an event message.

· If the value spikes over 1.0 amps then the spacecraft will execute a load shed.  

· The almost total lack of ability to play back engineering data prevents the FOT from being able to properly analyze this anomaly.

· FOT has no knowledge of the actual value of the current unless the event occurs during a scheduled ground contact.

· A violation of the first limit could indicate the value to be anywhere from 0.2 amps to 0.99 amps.

Mission Planning
Installation status of JPL’s Automated Scheduling and Planning ENvironment (ASPEN) in the MOC
· Computer hardware required to install ASPEN in the control center has arrived on base.

· FOT will begin System Administration of the new hardware and then explore installing the ASPEN ground system.
Flight Dynamics
Special image request

· The Mission Science Office has requested a data collection event over the North Pole.

· FOT has already designed the necessary attitude maneuvers to slew over the target.

· Targets of opportunity will be discussed with the project to execute this sequence.

· FOT will build a command load that includes the attitude and instrument commanding.

· This load will be up-linked to the spacecraft and executed as an Absolute Time Sequence during an S-Band contact to capture all possible telemetry.

· Due to the unusual attitude that EO-1 will be in while taking the image, FOT intends to monitor temperature values during the event to ensure the safety of instruments.   
Data Analysis and Trending
Integrated Trending and Plotting System (ITPS) development for EO-1

· Database files have been generated and delivered to the developers by the FOT.

· Raw data files for playback have been supplied to the developers for testing.

System Administration
No change from last report.
Operational Enhancements Requests and Discrepancy Reports
No change from last report.
GROUND AND SPACE NETWORK

Station Downtime
No major problems to report.
Operational Discrepancies
No major problems to report.
Support Enhancements  
No major problems to report.
Upcoming Events
A pair of MLT control inclination maneuvers is scheduled to be executed.

· First maneuver will be performed on May 27, 2008.

· Second maneuver will be performed on May 29, 2008.
A meeting will be held to discuss all IT related issues with the project management.

· Meeting will be on Friday, April 25 at 1:30pm in Building 14, Room W212.

Imagery Status
Scenes and Engineering Cals planned for week of April 17 – April 23, 2008              102
Total scenes and engineering calibrations planned for entire mission – approximately 39,446
Total Scenes:  ALI scenes in the Level 0 archive            35,843 (as of April 23, 2008)

                        Hyperion scenes in the Level 0 archive    35,589
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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